Author’s response to reviews

Title: Immunomodulatory factors gene polymorphisms in chronic periodontitis: An Overview

Authors:

Zahra Heidari (histology_iri@yahoo.com)
Bita Moudi (bita.moudi@yahoo.com)
Hamidreza Mahmoudzadeh-Sagheb (histology@ymail.com)

Version: 1 Date: 09 Jun 2018

Author’s response to reviews:

Dear Editor in chief

The manuscript was edited based on referees, comments as follows:

some textual overlap with other previously published works, were removed.

also, the Abstract restructured into the following sections: Background, Main Body, Conclusion.